Author’s response to reviews

Title: A comparison of licensed and un-licensed artisanal and small-scale gold miners (ASGM) in terms of socio-demographics, work profiles, and injury rates

Authors:

Benedict Calys-Tagoe (calys75@hotmail.com)

Edith Clarke (edith.clarke@ghsmail.org)

Tom Robins (trobins@umich.edu)

Niladri Basu (niladri.basu@mcgill.ca)

Version: 5 Date: 29 Sep 2017

Author’s response to reviews:

1. Please provide a 'Conclusions' section heading for the concluding paragraph of your manuscript.

Response: Section heading has been duly provided for the concluding paragraph.

2. Please rename your 'Ethical approval' section 'Ethics approval and consent to participate'

Response: Recommended change has been effected.

3. Please refer specifically to each author and their contributions in your 'Authors' Contributions' section. Please use initials to refer to each author's contribution in this section, for example: "FC analyzed and interpreted the patient data regarding the hematological disease and the transplant. RH performed the histological examination of the kidney, and was a major contributor in writing the manuscript. All authors read and approved the final manuscript."

Response: Reference has been made to each author and their contributions under the 'Authors contributions' section.